Author's response to reviews

Title: "Bong lung" in Cystic Fibrosis: a case report

Authors:

Zoe Gao (zoegao@gmail.com)
Richard Wood-Baker (Richard.WoodBaker@utas.edu.au)
Robin Harle (rharle@netspace.net.au)
Kon Muller (Kon.Muller@utas.edu.au)
Jenny Hauser (jenny.busch@dhhs.tas.gov.au)
David W Reid (D.E.C.Reid@utas.edu.au)

Version: 7 Date: 21 September 2010

Author's response to reviews: see over
Dear Editor

I have addressed all the queries and revised the manuscript as suggested,

Regards

David

Dr David Reid

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution
DONE

Abstract: please ensure that the Abstract in the manuscript matches exactly what is in the submission system, taking into account changes made with track changes
DONE

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
DONE

Tracking/coloured text: Please remove all tracking or coloured text from the manuscript.
DONE

Affiliations: Please do not use full capitals for country names

Qualifications: Please remove the authors' qualifications (e.g. PhD MD) or job titles from the manuscript and submissions system
DONE

Header: Please remove the header from the manuscript
DONE

Abbreviations: Please format these in to a paragraph.
DONE

Structure: Please ensure the Competing interests section precedes the Consent section.
DONE

Acknowledgments: Please remove this section if it is not applicable
DONE

Authors' contributions: Please remove the quotation marks from this section
DONE
Figures: Please move the figure legend section to after the reference list and rename it 'Figure legends'.

DONE